Author's response to reviews

Title: Early nasogastric tube feeding in optimising treatment for hyperemesis gravidarum: the MOTHER randomised controlled trial (Maternal and Offspring outcomes after Treatment of HyperEmesis by Refeeding)

Authors:

Iris J Grooten (i.j.grooten@amc.uva.nl)
Ben Willem J Mol (b.w.mol@amc.uva.nl)
Joris AM van der Post (j.a.vanderpost@amc.uva.nl)
Carrie Ris-Stalpers (c.ris@amc.uva.nl)
Marjolein Kok (m.kok@amc.uva.nl)
Joke MJ Bais (j.m.j.bais@mca.nl)
Caroline J Bax (c.bax@vumc.nl)
Johannes J Duvekot (j.j.duvekot@erasmusmc.nl)
Henk A Bremer (h.bremer@rdgg.nl)
Martina M Porath (m.porath@mmc.nl)
Wieteke M Heidema (wieteke.heidema@radboudumc.nl)
Kitty WM Bloemenkamp (k.w.m.bloemenkamp@lumc.nl)
Hubertina CJ Scheepers (hci.scheepers@mumc.nl)
Maureen TM Franssen (m.t.m.franssen@umcg.nl)
Martijn A Oudijk (m.a.oudijk-3@umcutrecht.nl)
Tessa J Roseboom (t.j.roseboom@amc.uva.nl)
Rebecca C Painter (r.c.painter@amc.uva.nl)

Version: 8 Date: 7 January 2016

Author's response to reviews: see over
Dear Editor,

As requested, we have uploaded a clean copy of the manuscript and edited the section on Authors’ contributions. Each author is now referred to by their initials and the role played by each has been clarified. There were no further reviewer comments to address.

Hoping to have addressed all questions sufficiently,

On behalf of all co-authors,

Iris Grooten, MD